Author's response to reviews

Title: Suboptimal blood pressure control in chronic kidney disease stage 3: baseline data from a cohort study in primary care

Authors:

Simon DS Fraser (s.fraser@soton.ac.uk)
Paul J Roderick (pjr@soton.ac.uk)
Natasha J McIntyre (cwm66@btinternet.com)
Scott Harris (sharris@soton.ac.uk)
Christopher W McIntyre (chris.mcintyre@nottingham.ac.uk)
Richard J Fluck (richard.fluck@nhs.net)
Maarten W Taal (maarten.taal1@nhs.net)

Version: 4 Date: 4 June 2013

Author's response to reviews:

June 2013
Submission to: BMC Family Practice
Corresponding author: Dr Simon DS Fraser.

Author address:
Public Health Sciences and Medical Statistics,
Level C, South Academic Block
Southampton General Hospital
Tremona Road,
Southampton
SO16 6YD
Tel: 023 80 796530
Fax: 023 80 796529
Author email address: s.fraser@soton.ac.uk

Dear Dr Foote,

Ref: MS: 7264111299318339

Research article: Suboptimal blood pressure control in chronic kidney disease
Thank you very much for your email confirming acceptance of our paper subject to adding a conflicts of interest statement. We have now added a statement that declares that none of the authors have a competing interest in the publication of this paper. This research was supported by grants from Kidney Research UK, and the British Renal Society and an educational grant from Roche, but none of the authors gain financially from the publication of this paper. See the response to the specific issues below

Kind regards

Dr Simon Fraser
On behalf of the authors

Comment

Authors are required to complete a declaration of competing interests. All competing interests that are declared will be listed at the end of published articles. Where an author gives no competing interests, the listing will read 'The author(s) declare that they have no competing interests'.

Response

A statement to this effect has been added to the manuscript under the section 'Competing interests and acknowledgements'

Comment

When completing your declaration, please consider the following questions:

Financial competing interests

In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

Response

No

Comment
Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

Response
No

Comment

Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

Response
No

Comment

Do you have any other financial competing interests? If so, please specify.

Response
No

Comment

Non-financial competing interests

Are there any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

Response
No